Author's response to reviews

Title: Pediatric non alcoholic fatty liver disease: old and new concepts on development, progression, metabolic insight and potential treatment targets

Authors:

Valentina Giorgio (valentagio@yahoo.it)
Federica Prono (fede_e.1990@hotmail.it)
Francesca Graziano (fra26_11@hotmail.it)
Valerio Nobili (nobili66@yahoo.it)

Version: 4 Date: 1 March 2013

Author's response to reviews: see over
Dear Editor,

As agreed with Sabina Alam, Editor of BMC Medicine, we are submitting the manuscript entitled “Pediatric non alcoholic fatty liver disease: old and new concepts on development, progression, metabolic insight and potential treatment targets” by Giorgio V, Prono F, Graziano F and Nobili V.

Don’t hesitate to contact us in case of any other need of changes or question.

We hereby state that the paper is original and that is not under consideration elsewhere for similar purposes. Each author listed on the manuscript has seen and approved the submission of this version of the manuscript and takes full responsibility for the manuscript.

Conflict of interest: none.

In case of acceptance of the manuscript the copyright will be transferred to BMC Medicine.

Thank you in advance for your interest.

Best regards,

Valentina Giorgio, MD

**Corresponding Author:**

Dr. Valentina Giorgio  
Bambino Gesù Children Hospital  
Piazza Sant’Onofrio 4, 00100, Roma, Italy  
Tel. +39 (0)6-6859 2243, FAX +39 (0)6-6859 3889  
E-mail address: valentagio@yahoo.it